Author's response to reviews

Title: Translation and Validation of the Arab Version of the Late-Life Function and Disability Instrument: A Cross Sectional Study

Authors:

Michal Elboim-Gabyzon (michal.elboim@gmail.com)
Maayan Agmon (agmon.mn@gmail.com)
Faisal Azaiza (azaiza@univ.haifa.ac.il)
Yocheved Laufer (yocheved@research.haifa.ac.il)

Version: 5 Date: 23 March 2015

Author's response to reviews: see over
Submission of revised manuscript: MS: 7254086901449241

Dear Dr. Tomohiro Okura,

We were are very grateful about the decision to accept our manuscript.

We added an Acknowledgement Section and the authors reports no conflict of interest.

Tables were included at the end of the main manuscript and removed from the additional files.

Additionally the table legends were inserted after the references.

Sincerely,

Dr. Michal Elboim Gabizon.